Author's response to reviews

**Title:** Experience of Nurses with using eHealth in GilgitBaltistan, Pakistan: a qualitative study in primary and secondary healthcare.

**Authors:**

Saleema Gulzar (saleema.gulzar@aku.edu)
Shariq Khoja (shariq.khoja@aku.edu)
Afroz Sajwani (afroz.sajwani@gmail.com)

**Version:** 4 **Date:** 28 June 2012

**Author's response to reviews:** see over
June 28, 2012

*BMC Nursing*
BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

**Subject: Revision of manuscript entitled ‘Experience of Nurses with using eHealth in Gilgit Baltistan, Pakistan: a qualitative study in primary and secondary healthcare’**

Dear Elizabeth H Anderson,

We have amended the ethics and consent statement as suggested to:

- Ethics approval was approved from Aga Khan University Ethical Review Committee.
- A written consent for participation in the study was signed by each participant.

Look forward for your positive response.

Sincerely,

Shariq Khoja MBBS. MSc. PhD (e-Health)
Director- AKDN eHealth Resource Centre
The Aga Khan University
3rd Parklands Avenue
Nairobi, PO Box 30270-00100
Kenya
Phone: 254 20 3662236
Fax: 254 20 3744956
Email: shariq.khoja@aku.edu